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ISR Chamber Tests Status

• Gas installation inside tunnel completed but no progress   
on outside gas shelter => still on pre-mixed bottles 

• New set of trigger counters installed under the chamber 
operational => 53 Hz trigger rate (old trigger => 4 Hz)

• No progress on Ethernet + phone upgrade, yet another 
option (wireless hook up) obscures real issue => who pays

• MB2 chambers repaired (HV) but still noisy wires

• Legnaro scaler system would allow to make fast noise 
measurements on chamber stacks but fixes are difficult at 
the ISR

• List of Urgent items not shrinking……
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Urgently Needed

• Final gas (water) connectors 

• RPC docking pads (time consuming operation at 
ISR, should be done at sites)

• HV cables, still 8days away from definition ???

• Analysis programs, strategy (large volume of 
channels to look at)

• Standardization of QC parameters, definition of 
good chamber in terms of:

- leak rate, dead channels, noisy channels,….

• Chamber handling tools

• Second HV system…….


